Author's response to reviews

Title: Design considerations of a randomized clinical trial on a cognitive behavioural intervention using communication and information technologies for managing chronic low back pain

Authors:

Julio Domenech (julio.domenech@uch.ceu.es)
Rosa Baños (rosa.banos@uv.es)
Lourdes Peñalver (upeba1965@gmail.com)
Azucena garcia-Palacios (azucena@uji.es)
Rocio Herrero (rherrero@uji.es)
Aida Ezzedine (a.ezzeddine@comv.es)
Monica Martinez-Diaz (monitxa10@hotmail.com)
Jaime Horta (jaimehorta@hotmail.com)
Javier Ballester (jballestersalva@hotmail.com)
Cristina Botella (botella@uji.es)

Version: 2 Date: 21 March 2013

Author's response to reviews:

1- We have added the following sentence in Methods: "Written informed consent will be obtained from all participants"

2- The authors’ contribution text has been expanded to detail each author contribution

3- We enclose a copy of the certificate of the Ethics Committee and resolution of the grant by the Carlos III Institute of the Spanish Government. For both we include the translation to English